Author's response to reviews

Title: High incidence of prostate cancer metastasis in Afro-Brazilian men with low educational levels: a retrospective observational study

Authors:

Alexandre BC Souza (alexbcsouza@hotmail.com)
Hugo G Guedes (hugogguedes@yahoo.com.br)
Victor CB Oliveira (victorcarbone.med@gmail.com)
Fábio A Araujo (bioaires@gmail.com)
Carlos CO Ramos (raijrn2011@gmail.com)
Karina CP Medeiros (karinapm@yahoo.com)
Raimundo F Araujo Jr (araujojr@cb.ufrn.br)

Version: 6 Date: 11 November 2012

Author's response to reviews:

Dear Editor,

We also corrected some language corrections with EDANZ GROUP,
Thanks for the suggestion,

Alexandre Barbosa.